Author's response to reviews

Title: The Value of Radiographic Findings for the Progression of Pandemic 2009 Influenza A/H1N1 Virus Infection

Authors:

Takanori Funaki (funaki-t@ncchd.go.jp)
Kensuke Shoji (shoji-k@ncchd.go.jp)
Nobuyuki Yotani (yotani-n@ncchd.go.jp)
Tomohiro Katsuta (jesus@pf6.so-net.ne.jp)
Osamu Miyazaki (miyazaki-o@ncchd.go.jp)
Shunsuke Nosaka (nosaka-s@ncchd.go.jp)
Hidekazu Masaki (masaki-h@ncchd.go.jp)
Akihiko Saitoh (asaitoh@med.niigata-u.ac.jp)

Version: 2 Date: 29 May 2012

Author's response to reviews: see over
May, 29, 2012

Philippa Harris
Executive Editor
BMC infectious diseases
BioMed Central

Dear Dr. Philippa Harris,

The manuscript, “The Value of Radiographic Findings for the Progression of Pandemic 2009 A/H1N1 Influenza in Children” is submitted to the BMC infectious diseases. To our knowledge, this article is the first report demonstrating the value of initial chest radiography to predict disease progression of children infected with pandemic 2009 A/H1N1 influenza. Our results showed patients with consolidation on chest radiograph tend to admit more frequently in pediatric intensive care unit and required mechanical ventilation.

I revised my manuscript about ethics statement, as you requested before. I described ethics statement in the Informed consent of Methods section of the manuscript. Please be sure to make a note of these changes.

All authors have seen and approve of the content and have contributed significantly to the work. It has not been published in this, or a substantially similar form, accepted for publication elsewhere, is it under consideration by another publication.

We hope you will find our manuscript acceptable for publication in the BMC infectious diseases. Thank you for your consideration.

Sincerely,

Takanori Funaki, M.D.
Fellow,
Division of Pediatric Infectious Disease, Department of Medical Specialties
National Center for Child Health and Development
Tokyo, Japan